Author's response to reviews

Title: Design of a school-based randomized trial to reduce smoking among 13 to 15-year olds, the X:IT study

Authors:

Anette Andersen (anan@niph.dk)
Lotus S Bast (loni@niph.dk)
Lene W Ringgaard (lener@cancer.dk)
Louise Wohlebe (lwo@cancer.dk)
Poul D Jensen (pajensen@hjerterforeningen.dk)
Maria Svendsen (msve@niph.dk)
Peter Dalum (pd@cancer.dk)
Pernille Due (pdu@niph.dk)

Version: 5
Date: 15 May 2014

Author's response to reviews:

Dear editor and reviewers,

Thank you for reading and commenting our paper again. We have now carefully checked the manuscript for spelling and grammar and it has been proof read by professionals.

Kind regards,
Anette Andersen